Author's response to reviews

Title: Autophagy and apoptosis-related genes in chronic liver disease and hepatocellular carcinoma

Authors:

Andromachi Kotsafti (andromachi.kotsafti@unipd.it)
Fabio Farinati (fabio.farinati@unipd.it)
Romilda Cardin (romilda.cardin@unipd.it)
Umberto Cillo (cillo@unipd.it)
Donato Nitti (donato.nitti@unipd.it)
Marina Bortolami (marina.bortolami@unipd.it)

Version: 2 Date: 13 April 2012

Author's response to reviews: see over
Dear Editor,

enclosed please find the complete copy of a paper entitled “Autophagy and apoptosis-related genes in chronic liver disease and hepatocellular carcinoma” that we hope is eligible for publication in the Journal of BMC Gastroenterology.

We state that the material submitted has not been previously published, and is not under consideration elsewhere.

Looking forward to hearing from you

Yours sincerely,

Marina Bortolami

Dr. Marina Bortolami, PhD
Department of Surgical, Oncological and Gastroenterological Sciences
Division of Gastroenterology
Via Giustiniani, 2
35128 Padova (Italy)
Telephone ++39 049 8212812
Fax ++39 049 8760820
E-mail: marina.bortolami@unipd.it